Author's response to reviews

Title: Evolving epidemiology and antimicrobial resistance in spontaneous bacterial peritonitis: a two-year observational study

Authors:

lionel Piroth (lionel.piroth@chu-dijon.fr)
andré Péchinot (andre.pechinot@chu-dijon.fr)
vincent Di Martino (ydimartino@chu-besancon.fr)
Yves Hansmann (Yves.Hansmann@chru-strasbourg.fr)
Alain Putot (alain.putot@chu-dijon.fr)
Isabelle Patry (ipatry@chu-besancon.fr)
Tahar Hadou (t.hadou@chu-nancy.fr)
Benoit Jaulhac (jaulhac@unistra.fr)
Catherine Chirouze (cchirouze@chu-besancon.fr)
Christian Rabaud (c.rabaud@chu-nancy.fr)
Alain Lozniewski (a.lozniewski@chu-nancy.fr)
Catherine Neuwirth (catherine.neuwirth@chu-dijon.fr)
Pascal Chavanet (pascal.chavanet@chu-dijon.fr)
Anne Minello (anne.minello@chu-dijon.fr)

Version: 4 Date: 6 May 2014

Author's response to reviews: see over
I made no change following the point 2 since:

1) the definition used is the official one which defines multidrug resistant bacteria 
2) the antibiotic classes of interest will differ according to the bacteria considered, 
3) so adding examples of the different classes of antibiotics will not add useful information and will make the text more difficult to read and understand 

Let me know if I am wrong

The two other points have been modified accordingly to the recommendations